Author's response to reviews

Title: The precarious supply of physical therapists across Canada: exploring national trends in health human resources (1991 to 2005)

Authors:

Michel D Landry (mike.landry@utoronto.ca)
Thomas C Ricketts (ricketts@schsru.unc.edu)
Molly C Verrier (m.verrier@utoronto.ca)

Version: 3  Date: 22 August 2007

Author's response to reviews:

Following editing and formatting